Author’s response to reviews

Title: Service use, unmet needs, and barriers to services among adolescents and young adults with autism spectrum disorder in Poland

Authors:

Mateusz Platos (mateusz.platos@gmail.com)
Ewa Pisula (ewa.pisula@psych.uw.edu.pl)

Version: 3 Date: 13 Aug 2019

Author’s response to reviews:

Dear Editor Ehsan Zarei,

Thank you for all your suggestions. We have included the ‘Abbreviations’ section in the manuscript and double-checked if all the abbreviations were defined in the text. We have also corrected the ‘Ethics consent’ section in line with your comment. Finally, we made sure it is a clear and final version of the manuscript, with all the tables being cited in the text. Please note that Table 1 is more than a page long so it is attached at the end of the article and a place where it is supposed to appear is indicated in the text. Hope these revisions meet your expectations.

Kind regards,

Mateusz Platos